

# G OPEN ACCESS

**Citation:** Sarma DK, Kumar M, Balabaskaran Nina P, Balasubramani K, Pramanik M, Kutum R, et al. (2022) An assessment of remotely sensed environmental variables on Dengue epidemiology in Central India. PLoS Negl Trop Dis 16(10): e0010859. <u>https://doi.org/10.1371/journal.</u> <u>pntd.0010859</u>

**Editor:** Amy C. Morrison, University of California Davis, UNITED STATES

Received: September 2, 2021

Accepted: September 30, 2022

Published: October 17, 2022

**Copyright:** © 2022 Sarma et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Climate data are publicly available for researchers at https://power. larc.nasa.gov/data-access-viewer/. The epidemiological datasets along with codes for analyses are available in the figshare repository under the URL: https://figshare.com/articles/ dataset/An\_assessment\_of\_remotely\_sensed\_ environmental\_variables\_on\_Dengue\_ epidemiology\_in\_Central\_India/20765668 (https:// doi.org/10.6084/m9.figshare.20765668). The numerical data that were used to generate graphs, **RESEARCH ARTICLE** 

# An assessment of remotely sensed environmental variables on Dengue epidemiology in Central India

Devojit Kumar Sarma<sup>1</sup><sup>\*</sup>, Manoj Kumar<sup>1</sup><sup>°</sup>, Praveen Balabaskaran Nina<sup>2,3</sup><sup>°</sup>, Karuppusamy Balasubramani<sup>4</sup><sup>°</sup>, Malay Pramanik<sup>5</sup>, Rintu Kutum<sup>6,7</sup>, Swasti Shubham<sup>1</sup>, Deepanker Das<sup>1</sup>, Manoj Kumawat<sup>1</sup>, Vinod Verma<sup>8</sup><sup>°</sup>, Jigyasa Dhurve<sup>1</sup>, Sekar Leo George<sup>4</sup>, Alangar Balasundreshwaran<sup>4</sup>, Anil Prakash<sup>1</sup>\*, Rajnarayan R. Tiwari<sup>1</sup>

 ICMR- National Institute for Research in Environmental Health, Bhopal Bypass Road, Bhouri, Bhopal, Madhya Pradesh, India, 2 Department of Epidemiology and Public Health, Central University of Tamil Nadu, Thiruvarur, Tamil Nadu, India, 3 Department of Public Health and Community Medicine, Central University of Kerala, Kasaragod, Kerala, India, 4 Department of Geography, School of Earth Sciences, Central University of Tamil Nadu, Thiruvarur, Tamil Nadu, India, 5 Urban Innovation and Sustainability Program, Department of Development and Sustainability, Asian Institute of Technology, Klong Luang, Pathumthani, Thailand,
 Department of Computer Science, Ashoka University, Sonipat, Haryana, India, 7 Trivedi School of Biosciences, Ashoka University, 8 Stem Cell Research Centre, Department of Hematology, Sanjay Gandhi Post-Graduate Institute of Medical Sciences, Lucknow, Uttar Pradesh, India

These authors contributed equally to this work.
 \* dkbiotek@gmail.com (DKS); anilprakashin@yahoo.co.in (AP)

# Abstract

In recent decades, dengue has been expanding rapidly in the tropical cities. Even though environmental factors and landscape features profoundly impact dengue vector abundance and disease epidemiology, significant gaps exist in understanding the role of local environmental heterogeneity on dengue epidemiology in India. In this study, we assessed the role of remotely sensed climatic factors (rainfall, temperature and humidity) and landscape variables (land use pattern, vegetation and built up density) on dengue incidence (2012-2019) in Bhopal city, Central India. Dengue hotspots in the city were assessed through geographical information system based spatial statistics. Dengue incidence increased from 0.59 cases in 2012 to 9.11 cases in 2019 per 10,000 inhabitants, and wards located in Southern Bhopal were found to be dengue hotspots. Distributed lag non-linear model combined with quasi Poisson regression was used to assess the exposure-response association, relative risk (RR), and delayed effects of environmental factors on dengue incidence. The analysis revealed a non-linear relationship between meteorological variables and dengue cases. The model shows that the risk of dengue cases increases with increasing mean temperature, rainfall and absolute humidity. The highest RR of dengue cases (~2.0) was observed for absolute humidity  $>60 \text{ g/m}^3$  with a 5–15 week lag. Rapid urbanization assessed by an increase in the built-up area (a 9.1% increase in 2020 compared to 2014) could also be a key factor driving dengue incidence in Bhopal city. The study sheds important insight into the synergistic effects of both the landscape and climatic factors on the transmission dynamics of dengue. Furthermore, the study provides key baseline information on the climatic

# tables was uploaded as <u>S1 Data</u> under <u>Supplementary Information</u>.

Funding: The study was funded by Indian Council of Medical Council, New Delhi (<u>https://main.icmr.</u> <u>nic.in/</u>) to DKS (award number 6/9-7(208)/2019-ECD-II), and by Department of Health Research, Government of India (<u>https://dhr.gov.in/</u>) to DKS, PBN and BK (award number R.11013/11/2021-GIA/HR). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

variables that can be used in the micro-level dengue prediction models in Bhopal and other cities with similar climatic conditions.

## Author summary

Dengue, a viral disease transmitted by infected Aedes mosquitoes, is a major public health concern globally. In addition to its increased incidence in recent years, dengue is also spreading to new geographical regions. Local environmental factors are known to modify the mosquito vector density that directly impacts dengue virus transmission. Understanding the influence of environmental factors (meteorological conditions and landscape features) on dengue epidemiology in local settings is important for focused dengue intervention. Here, by utilizing dengue incidence and remotely sensed environmental data from 2012–2019, we have assessed the role of environmental factors in driving dengue virus transmission in the city of Bhopal in Central India. During the study period, a 14.5 fold increase in dengue incidence was observed in Bhopal city, which is way higher than the 2.3 fold increase reported at the national level. The risk of dengue virus transmission was higher with higher temperature and absolute humidity. An increase in built-up area, a proxy for urbanization, was found to be another predictor of increased dengue incidence in Bhopal. These findings can provide a stepping-stone for the development of dengue prediction models and the identification of dengue hotspots in order to improve vector control of this disease in cities with similar environmental conditions.

# Introduction

Dengue is a serious mosquito-borne viral infection that affects more than half of the world's population. Globally, a 30 fold increase in dengue cases has been observed in the last five decades [1,2]. In recent years, the highest dengue incidence has been reported in South Asia, and is followed by Southeast Asia and tropical Latin America [3]. Climate change and urbanization are expected to facilitate the geographical expansion of dengue further, leading to a severe public health burden [4]. With an estimated annual global morbidity of >104 million and >40,000 deaths in 2020, dengue attributed Disability Adjusted Life Years have increased by ~ 107% in 195 countries [3,5,6].

Ever since the first recorded dengue outbreak in 1963 from Calcutta city in India [7], many outbreaks at various time periods have been reported from different parts of the country [8,9]. Dengue virus is primarily transmitted by *Aedes aegypti* and *Ae. albopictus* mosquitoes. India has witnessed an increase in dengue cases since 2001 with all four serotypes [2,10,11]. Initially, dengue was more predominant in urban and semi-urban areas in India. However, dengue has recently spread to rural areas, and could affect the entire Indian population [11–13]. During 2010–2014, the dengue incidence in India increased by more than 5 folds (34.81 per million) as compared to 1998–2009 (6.34 per million) [2]. Even though dengue is a notifiable disease in India, many studies suggest it is grossly under-reported, and the actual burden of dengue in the country is poorly quantified [14,15]. In 2019, a total of 157,315 cases of dengue and 166 deaths related to dengue were reported in India [16]. On the other hand, the Global Burden of Disease 2017 study reported an age-standardized dengue incidence rate of 4072.9 per 100,000 inhabitants in India which is one of the highest globally [3]. A nationwide serosurvey carried out in 2017–18 estimated 48.7% of the Indian population were seropositive for dengue [15].

Dengue virus transmission is highly influenced by meteorological factors such as temperature, rainfall and relative humidity (RH) [<u>17</u>]. Warm temperatures provide favorable conditions for the growth and development of vector mosquitoes and affect the length of the gonotrophic cycle, the extrinsic incubation period (EIP) of the virus within *Aedes* mosquitoes, and the basic reproduction number ( $R_0$ ), thereby influencing the transmission dynamics of the disease [<u>18-26</u>]. The diurnal temperature range also influences the infectivity of the dengue virus in *Ae. aegypti* [<u>27</u>]. Similarly, rainfall provides habitat for mosquito breeding and increases the abundance of vector mosquitoes [<u>28</u>]. High humidity also favors the survival and biting of adult vector mosquitoes [<u>29,30</u>].

Dengue transmission is also influenced by anthropogenic and environmental factors such as the degree of urbanization, human mobility, land use change, and vegetation [31–33]. An increase in urbanization has led to heterogeneous socio-economic and environmental conditions that facilitate dengue transmission [34]. Increased population density along with poor socio-economic conditions and distressed environmental hygiene provide a suitable environment for vector breeding and increased human-vector contact and dengue transmission [34– 38]. In the past few decades, India has been undergoing rapid urbanization, and as of 2018, 34% of India's population were living in urban areas compared to 11% in 1901 [39]. Rapid urbanization with concomitant changes in vegetation cover and land use types will result in urban heat islands (warmer temperatures than nearby rural areas) that can significantly impact urban micro-climates [40]. This in turn can modify the local vector populations, increase hostvector contact and alter the transmission dynamics of dengue virus [41–44].

Even though most Indian cities experience a tropical or subtropical climate, local variations in topography and land use affect micro-climatic conditions and local dengue epidemiology. This environmental heterogeneity has also been documented by differences in EIP, an important entomological parameter to measure dengue epidemiology in different parts of the country [2]. Due to this environmental heterogeneity, a common dengue disease model for early warning and outbreak prediction may not be applicable to the entire country [2,45]. An elegant study by Kakarla et al. [45] modeled the lagged effects of meteorological variables on dengue incidence in India. However, significant gaps persist, especially on our understanding on the role of local climatic and landscape features on dengue incidence at the city level. To understand the role of local climatic factors and urbanization on dengue incidence (2012–19) in Bhopal city in Central India, relevant remotely sensed datasets were analyzed and the impact of environmental variables on dengue incidence is detailed. The outcome of this study will greatly help to understand, predict, and control dengue transmission in endemic areas with similar eco-climatic conditions.

### Materials and methods

#### Study area

Bhopal (23.25°N, 77.40°E), the capital city of Madhya Pradesh state in Central India, is located in the sub-tropical climate zone. It encompasses an area of 285.9 km<sup>2</sup> with an average altitude of 500m above mean sea level and possesses dry mixed deciduous forest cover. The city has a total population of 1.9 million [46], with a population density of 3,887 persons/km<sup>2</sup>. The city is divided into 85 wards, which act as administrative and health service delivery units (Fig 1). The topography of Bhopal city is uneven and characterized by a number of small hillocks and large water bodies. The city is surrounded by dense to open scrub forests, and the major land use types are sparse to high-density built-ups, water bodies, agricultural crop lands, and a range of fallow/waste lands. The climate is characterized by dry winter, hot summer, and humid monsoon seasons with an annual average temperature of 25°C (ranges from 15°C to



Fig 1. Location map and administrative boundaries of Bhopal city, Madhya Pradesh, India (Source of shapefile: <u>http://projects.datameet.org/Municipal\_Spatial\_Data/</u>).

https://doi.org/10.1371/journal.pntd.0010859.g001

 $40^{\circ}$ C), and rainfall of ~ 1200 mm. The summer season starts in March, peaks in May (exceeds  $40^{\circ}$ C) and continues till June. The monsoon season starts in late June and ends in late September. The winter season (November—February) peaks in January when the night temperature may drop close to the freezing point. The average RH ranges from 20% (summer) to 90% (monsoon).

### Data collection

**Collection of epidemiological data.** In India, dengue is diagnosed primarily based on clinical manifestations (high fever, headache, retro-orbital pain, myalgia, arthralgia, rash and hemorrhagic manifestations) and laboratory diagnosis [47]. Clinically suspected dengue cases are confirmed in the laboratory by IgM antibody-based method (MAC ELISA). A list of all confirmed dengue cases from January 2012 to December 2019 reported in Bhopal city was obtained from the District Malaria Office, Bhopal. This data contained date wise confirmed dengue cases along with their sex, age and address reported from government as well as private

hospitals. The locations of the reported cases were geotagged using Google Earth [48]. The dengue cases based on the date of reporting were entered in the SPSS software, and weekly and monthly dengue cases were aggregated for statistical and spatial analysis. The week wise dengue incidence data and their ward wise distribution has been deposited in Excel format in fig-share repository as an open access link (https://figshare.com/articles/dataset/An\_assessment\_of\_remotely\_sensed\_environmental\_variables\_on\_Dengue\_epidemiology\_in\_Central\_India/20765668). The study was approved by the Institutional Ethics Committee of ICMR-National Institute for Research in Environmental Health, Bhopal (NIREH/BPL/IEC/2018-19/3130 dated 18-03-2019).

**Collection of meteorological data.** Daily average values of rainfall (mm), maximum, minimum and mean temperature (°C), and RH (%), all measured at 2 metres from ground level, were collected from 1<sup>st</sup> January 2012 to 31<sup>st</sup> December 2019. The parameters were extracted from the National Aeronautics and Space Administration (NASA) Langley Research Center Prediction of Worldwide Energy Resource Project [49], which is based on Modern-Era Retrospective analysis for Research and Applications version 2 (MERRA-2) dataset merged with Goddard Earth Observing System Model (GEOS) version 5.12.4. Absolute humidity (AH, g/m<sup>3</sup>), was estimated using dry bulb temperature and RH at standard atmospheric pressure [50] using the following equation:

$$AH = 1000 \times (6.11 \times 10^{T_1} \times 100) / (T_c + 273.16) \times 461.5$$
<sup>(1)</sup>

Where, Tc is the dry bulb temperature (in our studies, Tc is the daily mean temperature), and

$$T_1 = 7.5 \times T_d / (237.7 + T_d)$$

Where, Td is the dew point temperature. Td is approximated from the equation below, based on dry bulb temperature and RH:

$$\begin{split} T_d &= (-430.22 + 237.7 \times \ln(\mathrm{E})) / (-\ln(\mathrm{E}) + 19.08), \text{ where}, \\ E &= RH \times E_{\mathrm{s}} / 100 \\ E_{\mathrm{s}} &= 6.11 \times 10^{T_2} \\ T_2 &= 7.5 \times T_{\mathrm{c}} / (237.7 + T_{\mathrm{c}}), \end{split}$$

Daily values of weather variables were converted to weekly values as simple arithmetic sums for rainfall and means for other parameters such as temperature, humidity etc., to assess the potential effect of seasonal variation on dengue dynamics.

**Extraction and processing of remote sensing data.** The multi-sensor satellite images were extracted through Google Earth Engine [48] and ArcGIS 10.2 software [51] to assess the change in land use/land cover (LULC) and vegetation index. Landsat-8 Operational Land Imagers (OLI) and Thermal Infra-Red Sensors (TIRS) images (S1 Table) were extracted for preparing LULC maps with nine LULC classes for two time periods (2014 and 2020) using the spectral signatures of multi-bands through the visual image interpretation approach. The accuracy of delineated LULC classes was assessed with the help of high-resolution true-colour Google Earth images and the LULC change transition matrix was prepared.

In addition to LULC, vegetation and built-up indices were prepared to assess the correlation between prominent land use variables and incident dengue cases. The Normalized Difference Vegetation Index (NDVI) is a numerical quantity derived from reflectance measured in the Near-Infrared (NIR) and Red spectral bands. NDVI provides consistent spatial and temporal comparisons of vegetation canopy greenness, a composite property of leaf area, chlorophyll and canopy structure [52,53]. The NDVI can be determined using the following equation.

$$NDVI = (NIR - Red)/(NIR + Red)$$
(2)

The Moderate Resolution Imaging Spectroradiometer (MODIS) NDVI data of the Terra satellite (MOD13Q1) products were used to estimate the monthly average NDVI value for Bhopal city in GEE platform for 96 months (January 2012 to December 2019) and was aggregated by ward using ArcGIS spatial statistics tool. Cross-correlations between the monthly averaged NDVI values and monthly cases were evaluated using Pearson's product-moment correlation coefficient ( $r^2$ ).

The LULC data downloaded from Copernicus Global Land Cover Service (CGLS) [54] was used to summarize ward wise annual built-up area from 2015 to 2019 using the ArcGIS 10.2 Spatial Analyst extension [51]. Furthermore, built-up areas (values  $\geq$ 1) were extracted, and ward wise built-up areas were computed using zonal statistics in the ArcGIS 10.2 software. The ward wise built-up density was calculated by dividing the built-up area by the total area of each ward. A stepwise backward multiple linear regression was used to evaluate the influence of these landscape variables (ward wise maximum, minimum, and average NDVI values along with built-up density) on dengue incidences.

**Statistical analyses.** Basic demographic characteristics such as numbers, median age, gender and age-group of dengue cases reported in Bhopal city during the study period (2012–2019) were analyzed year-wise. Continuous variables were summarized as mean and standard deviation when normally distributed, and median with inter-quartile range when non-normally distributed. Categorical variables were summarized as counts and proportions. For all analyses, p-value <0.05 was considered statistically significant. Time series of weekly dengue case count were plotted along with weekly average values of mean minimum and maximum temperature, mean RH, and total precipitation.

The annual average rate of change in the maximum, minimum and mean temperature over the study period in Bhopal city were estimated by the Mann-Kendall non-parametric test and Sen's slope estimate [55]. In this estimate, if "*n*" is nine or less, the test statistic S is applied. The absolute value of S is compared to the probabilities of the Mann-Kendall non-parametric test to study if there is a monotonic trend or not at the significance level of  $\alpha$ . A positive (or negative) value of S indicates an upward (or downward) trend. The Sen's slope estimate, Q is used to estimate the actual slope of the linear trend (change per year). An  $\alpha$  value of 0.1 is considered to be statistically significant [55].

**Spatial analysis.** Collected epidemiological and geographical data were converted into spatial layers and aggregated to the ward boundaries of Bhopal city using ArcGIS 10.2 software [51]. The ward wise cases from 2012 to 2019 were analyzed and normalized with the wards' total population (as per the 2011 census). The normalized cases per 10,000 inhabitants were used to obtain the spatial-temporal distribution of dengue cases in Bhopal city.

Spatial autocorrelation analysis was performed for case locations and ward wise case distribution using the spatial statistics tool in ArcGIS 10.2 software [51]. The optimized hotspot analysis through Getis-Ord Gi\*statistic was applied to date-wise locations of dengue cases by creating fishnet polygon mesh (cell size = 150 m) for aggregating the incidents. This spatial statistic works by looking at values of input variables for each case/ward location within the context of the neighbouring cases/wards. The local sum for each fishnet polygon and its neighbors is compared proportionally to the sum of all neighbouring cases/wards. The corresponding Z-score is calculated using Getis-Ord Gi\*statistic. To have a significant hotspot of dengue

incidence, it should have a location with high case density and adjacent areas with high dengue cases. The analysis found 506 statistically significant locations based on false discovery rate (FDR) correction for multiple testing and spatial dependence of dengue incidence. The locations/wards with high values of positive Z-scores (>1.96) are referred to as 'hotspots', and negative Z-scores are termed as 'coldspots'. If there are no hot or coldspots, the cases/wards would have a random spatial distribution.

**Estimation of lag effects.** To investigate the delayed impact of meteorological variables on the incidence of dengue cases, we employed two distinct approaches to assess this relationship. First, the association between weekly dengue incidence and meteorological variables at different time lags (0–30 weeks) was assessed using Pearson's cross-correlation analysis. Lag represents the time gap between exposure and clinical outcomes. The second part of the analysis was carried out using the distributed lag non-linear model (DLNM) of the "*dlnm*" package [56,57] available in the R software, v3.3.8 [58]. This model can examine non-linear relationships and addresses multi-collinearity issues by applying spline smoothing techniques or polynomial functions. It can handle lag effect is important in dengue transmission and climate fluctuations related to the duration of the mosquito life cycle and virus propagation [59]. Many studies have used this approach to evaluate non-linear relationships between climatic factors and mosquito-borne diseases [45,50,59,60].

DLNM combined with quasi-Poisson regression was used to calculate relative risk (RR) by estimating the effect of independent variables on the dependent variable. This model is represented for time series data by the following equation (Eq.3):

$$Y_t =$$
Quasi-Poisson  $(\mu_t), t = 1, 2, 3, ..., n$ 

$$Log(\mu_t) = \alpha + \sum_{l=1}^{L} \beta 1(T_{t,l}) + \sum_{l=1}^{L} \beta 2(R_{t,l}) + log(N_t) + s(t,\lambda) + Year + \varepsilon_t$$
(3)

Where, *t* is the week of the observation,  $Y_t$  denotes the observed dengue counts in week *t*,  $log(\mu_t)$  represents the logarithm of expected dengue cases in week *t*,  $\alpha$  is the model intercept;  $T_{t,b}$  and  $R_{t,b}$  are the matrices obtained by applying the DLNM to;  $\beta_1$ , and  $\beta_2$  are the vectors of coefficients for  $T_{t,l}$  and  $R_{t,b}$  *l* is the lag in weeks; *L* is the maximum lag (in our case it is 25 weeks);  $N_t$  is an offset to control for population using a linear function of time based on the 2001 and 2011 censuses population of Bhopal; *s*(*t*,  $\lambda$ ) is the natural cubic spline smoothing function of the calendar week and  $\varepsilon_t$  are the residuals added at specific lags to correct for partial autocorrelation. The mean values of each climatic variable (S2 Table) were used as reference to calculate RR.

Identification and removal of seasonality from time-series data will improve the modeling performance and will also result in a clean relationship between the input and output variables. Therefore, all climatic variables were tested for seasonality before using the DLNM modeling approach. This considerably decreases the collinearity problem while maintaining the model's interpretability. The MSTL (Multiple Seasonal-Trend decomposition using Loess) time series seasonal adjustment method was used using the "*mstl*" function in R [61], and seasonality adjusted data was extracted using the "*seasadj*" function included in the "*forecast*" package in R [62].

Similarly, the delayed effect of monthly average values of NDVI to dengue incidences was also assessed using the DLNM method.

The numerical data used in for figures are included in S1 Data.

Characteristics	2012	2013	2014	2015	2016	2017	2018	2019	Total
Number of cases	114	158	706	232	661	1094	615	1753	5333
Case Density/ 10,000 populations (95%CI)	0.59 (0.49– 0.71)	0.82 (0.70– 0.96)	3.67 (3.41– 3.95)	1.21 (1.06– 1.37)	3.44 (3.18- 3.71)	5.69 (5.36– 6.03)	3.20 (2.95- 3.46)	9.11 (8.69– 9.55)	27.72 (26.99– 28.47)
Median Age, years (IQR)	23 (11-33)	22 (18-33)	24 (16-35)	23 (17-34)	23 (17-35)	25 (18-39)	24 (16-33)	24 (18-35)	24 (14.5-32)
Sex, number (%)									
Male	71 (62.3)	112 (70.9)	441 (62.5)	164 (70.7)	429 (64.9)	712 (65.1)	381 (62.0)	1128 (64.3)	3438 (64.5)
Female	39 (34.2)	46 (29.1)	265 (37.5)	68 (29.3)	232 (35.1)	379 (34.6)	210 (34.1)	593 (33.8)	1832 (34.4)
Unspecified	4 (3.5)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	3 (0.3)	24 (3.9)	32 (1.8)	63 (1.2)
Age group (%)									
<5 Years	10 (8.8)	5 (3.2)	28 (4.0)	10 (4.3)	14 (2.1)	31 (2.8)	25 (4.1)	47 (2.7)	170 (3.2)
5-8 Years	8 (7.0)	10 (6.3)	32 (4.5)	11 (4.7)	22 (3.3)	22 (2.0)	38 (6.2)	48 (2.7)	191 (3.6)
9-17 Years	22 (19.3)	22 (13.9)	149 (21.1)	40 (17.2)	130 (19.7)	183 (16.7)	95 (15.4)	291 (16.6)	932 (17.5)
18-45 Years	52 (45.6)	102 (64.6)	409 (57.9)	144 (62.1)	411 (62.1)	669 (61.2)	372 (60.5)	1079 (61.6)	3238 (60.7)
> 45 Years	15 (13.2)	19 (12.0)	81 (11.5)	27 (11.6)	83 (12.6)	179 (16.4)	63 (10.2)	225 (12.8)	692 (13.0)
Unspecified	7 (6.1)	0 (0.0)	7 (1.0)	0 (0.0)	1 (0.2)	10 (0.9)	22 (3.6)	63 (3.6)	110 (2.1)

Table 1. Demographic characteristics of notified dengue cases in Bhopal city (2012-2019).

https://doi.org/10.1371/journal.pntd.0010859.t001

### Results

### Descriptive epidemiology of dengue in Bhopal

A total of 5,333 dengue cases were notified in Bhopal city during 2012–2019 (Table 1). A gradual increase in the dengue cases was observed over the study period. The case incidence varied from 0.59 cases/10,000 inhabitants (95% CI: 0.49–0.71) in 2012 to 9.11 cases/10,000 inhabitants (95% CI: 8.69–9.55) in 2019. The case incidence in Bhopal city was higher than the average of the entire state of Madhya Pradesh and India during the study period (S1 Fig). From 2012–2019, Bhopal witnessed a 14.5-fold increase in annual dengue incidence, while it was a 2.13-fold increase at the national level. More than 50% of the cases were males irrespective of the age groups (Table 1). The median age for the notified cases was 24 years (IQR: 14.5–32 years) with a male to female ratio of 1.8:1. The majority of the cases (60.7%) belonged to 18–45 years age group, and was followed by 9–17 years age group (17.5%). Dengue cases increased sharply during September to November, contributing to almost 82% of the yearly case load. The cases peaked in October, which accounts for 38% of the yearly cases (Fig 2A and 2B).



**Fig 2. Month wise distribution of dengue cases in Bhopal city (2012–2019).** A) Month wise percentage contribution of dengue cases. B) Month wise box plot distribution of dengue case incidence per 10,000 inhabitants (The box comprises 50% of the distribution, the line within the box is the median value, borderlines are the first and the third quartile, and the small circles are the outliers. February, March, June, and July had outliers).

https://doi.org/10.1371/journal.pntd.0010859.g002



Fig 3. Ward wise spatial distribution of dengue cases (2012–2019) indicating significant hotspots and coldspots in Bhopal city. The hotspots are shown in the map as shades of red to highlight the geographic areas where the clustering of dengue cases occurred. Similarly, coldspots with negative Z-scores are shown in blue to indicate areas where fewer-than-expected dengue cases were observed. (Source of shapefile: <a href="http://projects.datameet.org/Municipal\_Spatial\_Data/">http://projects.datameet.org/Municipal\_Spatial\_Data/</a>).

https://doi.org/10.1371/journal.pntd.0010859.g003

#### Ward wise distribution of dengue cases and dengue hotspots

Bhopal city witnessed a gradual expansion of dengue cases over the years (S1 Fig). Of the 85 wards, only 5 wards had >2.0 dengue cases/10,000 inhabitants in 2012, while in 2014 and 2019, 38 and 75 wards had >2.0 cases/10,000 inhabitants, respectively (S2 Fig), suggesting most of the city had sustained transmission of dengue. Spatial distribution showed the wards located in the southern and eastern parts of the city to be the significant dengue hotspots (Fig.3).

#### Effect of meteorological variables on dengue incidence

The weekly dengue cases, mean temperature, rainfall, RH and estimated AH from 1<sup>st</sup> January 2012 to 31<sup>st</sup> December 2019 in Bhopal city are presented in Fig 4. On an average, 13 cases per week and 664 cases per year were reported, with maximum cases occurring during weeks 38–45 (mid-September to mid-November) of the year. Descriptive statistics of all meteorological variables are shown in <u>S2 Table</u>. Even though the yearly peak of dengue cases varied slightly from year to year, most cases tended to occur following the monsoon season (Fig 4).

The weekly average of meteorological variables such as rainfall, humidity, diurnal temperature range, mean temperature, and AH exhibited significant Pearson cross-correlation



https://doi.org/10.1371/journal.pntd.0010859.g004

correlations with weakly dengue cases at different time lags (<u>S3 Fig</u>). A significant correlation between dengue and rainfall ( $r^2 = 0.510$ ), humidity ( $r^2 = 0.554$ ), diurnal temperature range ( $r^2 = 0.601$ ) and temperature ( $r^2 = 0.599$ ) were observed at 12, 8, 9 and >20 weeks lag respectively (<u>S3 Fig</u>).



Fig 5. Association of dengue incidence with different ranges of meteorological variables. A) RR by lag weeks at different rainfall (mm) levels. B) RR by lag weeks at different mean temperature (°C). C) RR by lag weeks at different AH ( $g/m^3$ ). D) RR by lag weeks at different RH (%). The solid green line is the estimated linear curve, with shaded region indicating its 95% confidence interval.

https://doi.org/10.1371/journal.pntd.0010859.g005

As no effect of seasonality was observed on the climatic variables based on the seasonality decomposition analysis (S4 Fig), the specific correlation of the climatic variables to dengue incidence was assessed by DLNM using the original data set. The delayed effect/lagged relationship between weekly meteorological variables (rainfall, mean temperature, AH) and dengue cases, analyzed using DLNM combined with quasi Poisson regression analysis, revealed a non-linear relationship (S5 Fig). When compared to the drier weeks (weekly rainfall <50mm), the wettest weeks (weekly rainfall >100mm) have a shorter lag period of 5 weeks with a RR of 1.08 (95% CI [0.99–1.17]) (Fig 5A). The association between mean temperature and dengue cases with a 25 week lag is shown in S5B Fig. The RR was 1.11 (95% CI [1.01–1.22]) at 35°C mean temperature with a 5–15 week-lag (peak at 10 weeks) compared to the reference value (25.9°C) (Fig 5B). Similarly, risk of dengue cases increases with increasing AH (S5C Fig). The RR of dengue cases was 1.42 (95% CI [1.07-1.88]), 1.73 (95% CI [1.11-2.71]) and 2.11 (95% CI [1.15-3.89]) for AH 60, 80 and 100 g/m<sup>3</sup> respectively with a 5–15 week lag (peaked at 10 weeks) (Fig 5C). The RH appears to follow the pattern of rainfall on RR of weekly dengue incidence; however, the DLNM model does not show any significant association (Figs 5D and <u>S5D</u>).

Land use area-2020												
Land use area— 2014	Classes	Agriculture	Dense scrubs	Open scrubs	Dense Built-up	Built-up with vegetation	Sparse Built-up	Structures/ Misc.	Water Body	Fallow/ Barren land	Land use area (ha) 2014	Change in Land use (ha) (2014–2020)
	Agriculture	45230.1	1.0	339.2	156.3	425.0	357.6	123.9			46633.1	-1401.5
	Dense scrubs	1.0	2077.9	11.1	0.1	1.1			3.9		2095.1	35.0
	Open scrubs	0.6	33.6	10372.5	94.7	77.5	45.9	94.6	14.9	17.9	10752.2	-9.1
	Dense Built- up				5939.1	0.1		1.3			5940.5	407.5
	Built-up with vegetation				81.9	5298.0					5379.9	421.9
	Sparse Built- up				75.9		2585.9				2661.7	358.5
	Structures/ Misc.							1452.2			1452.2	219.8
	Water Body		17.7	15.7					5953.5	4.6	5991.5	-16.2
	Fallow/Barren land			4.6			30.9		2.9	2021.2	2059.7	-16.0
	Land use area (ha) 2020	45231.6	2130.1	10743.1	6348.0	5801.8	3020.3	1672.0	5975.2	2043.7	82965.9	

https://doi.org/10.1371/journal.pntd.0010859.t002

#### Effect of urbanization on dengue incidence

The LULC change between 2014 and 2020 (Table 2) for Bhopal city (S6 Fig) revealed a 9.1% increase in the built-up classes and a 3% reduction in the agricultural land use. The crop land surrounding the city decreased considerably, and since 2013, greater than half of the agricultural land use (~765 ha) was converted to open scrub and built-up with vegetation. To understand the dominance of impervious cover, a yearly built-up area map was prepared for Bhopal city from 2015–2019, and built-up density was calculated for each ward. Since both vegetation cover and built-up density can be considered as a proxy for urbanization, we further computed the ward wise average minimum and maximum NDVI values along with built-up and population density. To better understand the role of these remotely sensed landscape variables on dengue incidence, a backward stepwise regression was applied. Minimum NDVI was the significant landscape variable influencing dengue incidence in Bhopal city (model  $r^2 = 0.035$ , F statistics = 16.293, p = 0.0001) (Table 3).

A slight increase in the monthly average NDVI and a corresponding increase in dengue cases were observed (S7A Fig). The cross-correlation of monthly average NDVI with dengue cases over the period was significantly positive ( $r^2 = 0.303$ ) in the same month (0 month lag period) (S3 Table). The strength of association slightly increased ( $r^2 = 0.339$ ) after a lag period of one month. Dengue incidence with RR of 1.42 (95% CI [0.99–1.84]) was observed at >4 months lag at average monthly NDVI value of 0.2 using the DLNM approach (S7B Fig).

### Discussion

Understanding the epidemiology of dengue in the context of environmental factors is important for its better management and control. In India, except for New Delhi [63], the role of climatic and landscape variables on dengue incidence and transmission at the city level has been poorly studied. We examined the relationship of various meteorological and landscape variables with dengue incidence in the city of Bhopal, Central India. Our study found increasing

Covariates	В	SE	Adjusted r <sup>2</sup>	t	Þ			
Model 1								
Intercept	2.304	1.413	0.037	1.631	0.104			
Built-up Density	-1.273	0.742		-1.716	0.087			
NDVI_Average	12.257	14.378		0.852	0.394			
NDVI_Maximum	-10.352	8.227		-1.258	0.209			
NDVI_Minimum	16.696	6.68		2.499	0.013			
Model 2								
Intercept	2.513	1.391	0.038	1.806	0.072			
Built-up Density	-1.25	0.741		-1.687	0.092			
NDVI_Maximum	-3.836	3.043		-1.261	0.208			
NDVI_Minimum	20.093	5.36		3.748	0			
Model 3			0.037					
Intercept	1.482	1.127		1.316	0.189			
Built-up Density	-0.917	0.693		-1.324	0.186			
NDVI_Minimum	15.906	4.211		3.778	0			
Model 3								
Intercept	0.626	0.923	0.035	0.678	0.498			
NDVI_Minimum	16.794	4.161		4.037	0			
Dependent variable: Dengue case density per 10000 inhabitants								
F-Statistics: 16.293								
<i>p</i> -value of F-statistics: 0.0001								

#### Table 3. Hierarchical multiple regression analysis with backward elimination of landscape variables on dengue incidence in Bhopal city.

B = Coefficient, SE = Standard error

https://doi.org/10.1371/journal.pntd.0010859.t003

mean temperature  $(30^{\circ}-35^{\circ}C)$  with 5–15 week-lag, low rainfall (20-40mm) with a longer lag (15–20 weeks), high rainfall (above 100mm) with a shorter lag (5–10 weeks) and high AH (above 60 g/m<sup>3</sup>) with a lag of 5–15 weeks to be the most significant meteorological variables associated with an increased dengue incidence in Bhopal city.

Temperature is a key climatic factor that affects both the life cycle of vector mosquitoes as well as the pathogen it harbors. Temperature influences the length of the gonotrophic cycle, larval development and the growth rate of mosquitoes [17]. An increase in temperature is associated with an increased dengue incidence due to faster viral replication rate, shortened EIP, and increased mosquito biting rate, leading to higher vectorial capacity and disease transmission [23,64,65]. DLNM shows a positive association of dengue with 30°-35°C at a lag of 5-15 weeks. This mean temperature range (30°-35°C) is usually maintained from mid-April to end of June in Bhopal (dry season). The desiccation resistant ability of Aedes eggs [66], and transovarian transmission capability of dengue virus through desiccated eggs [67] help them sustain the dry season which is followed by rainy season from mid-June allowing Aedes eggs to hatch, expand Aedes population and establish the dengue transmission cycle. In line with our findings, an association of elevated mean temperature with a long lagged dengue incidence (3–5 months) was also observed in Singapore [18,50,66,68], Nepal [69], and Thailand [29]. Similarly Lowe et al. also observed drought conditions positively associated with dengue infection at a longer lag up to 5 months in Barbados [70]. Interestingly, heat waves in Singapore were also associated with a reduction in dengue infections; a 1°C increase from 31 to 32°C in maximum temperature was associated with a 13.1% decrease in dengue incidence over 6 weeks [60]. In India, the risk of dengue transmission increases with an increase in mean temperature above

24°C, and the greatest transmission risk was reported to be at 30°C with a lag of 0–3 weeks [45]. India has different climatic zones and predicting the risk of dengue based on climatic variables at the national level may not correlate well in the country's diverse geographical settings. The average long-term temperature of major cities of India varies considerably due to its physiographic settings, for instance, the annual mean maximum temperature ranges 19°-41°C, 26°-32°C and 28°-37°C for Delhi, Bengaluru and Chennai cities respectively [71,72]. Clearly, the lag effect of temperature in different geographical settings in India needs to be ascertained to predict and plan future dengue control strategies.

Rainfall provides breeding habitats for mosquitoes, and Ae. aegypti density is an important variable and a predictor of dengue incidence in urban settings. Even though continuous heavy rainfall can washout breeding sites and negatively impact Ae. aegypti density [23,73,74], the extent of its effect is dependent on the intensity of rainfall, frequency of flush out events, size of the water holding containers, and larval age [75–77]. In Bhopal, rainfall and humidity increases from mid-June to September, and the mean temperature also rises above 30°C from April to June, and is accompanied by a rise in dengue cases; the surge in dengue cases is synchronized and correlated with delayed rainfall and temperature. Historically, in Bhopal, the number of dengue cases is low during the winter and dry period (January to May), and the peak is observed towards the end of the monsoon season (October). Many studies have reported a non-linear relationship between rainfall and dengue incidence. Increase in rainfall has shown to be positively associated with dengue, but with varied lag periods; in Malaysia, Sri Lanka and Brazil, the lag period is 4, 15–20 and 12 weeks respectively [78–80]. On a nationwide scale in India, the risk of dengue was highest at 60mm rainfall with a 12 week lag [45]. Dengue incidence in Bhopal is positively associated with high rainfall (100-120mm) and a shorter (5–10 weeks) lag period. Again, these findings underscore the importance of studying the meteorological variables in different climatic zones for efficient modeling of dengue transmission.

Most of the studies investigating the effect of environmental variables on dengue incidence have focused on temperature, rainfall and RH, while the effect of AH on dengue incidence has not been well described. We found the highest risk (RR: 1.42 to 2.11) of dengue incidence at a lag of 5–15 weeks above 60 g/m<sup>3</sup> AH in Bhopal. In Singapore, a similar effect of AH, rather than RH, on dengue incidence was observed [50], and it was found to be a better predictor of dengue incidence with respect to other climatic variables. In Guangzhou province, China, higher RH was negatively associated with dengue incidence [59]. Thus, along with the mean temperature (which provides a suitable environment for virus replication within vector mosquitoes), AH (which amplifies the virus transmission potential) may provide a more stable and better predictor for modeling dengue incidence, and must be considered together with other contributing factors for an effective dengue control policy recommendation.

In addition to climatic variables, urbanization and anthropogenic influences are also considered as the key drivers of dengue expansion [81]. *Ae. aegypti*, the highly dominant, adaptable and domesticated dengue vector, is highly anthropophilic, and prefers to live alongside humans, hence, urbanization and population growth provide ideal opportunities for the breeding and spread of *Ae. aegypti*. A positive temporal association of monthly average NDVI and dengue incidence at a lag of one month was observed using Pearson's cross-correlation. More specifically a higher risk of dengue was observed at a lower monthly average NDVI value with extended lag periods in comparison to higher monthly average NDVI values (<u>S7B and S7C</u> <u>Fig</u>). Similarly, the multiple regression model also suggests minimum NDVI as a significant risk factor for dengue incidence in the wards of Bhopal (<u>Table 3</u>). A higher dengue incidence rate was also found to be associated with low vegetation cover in São Paulo, Southeastern Brazil [82]. The maximum average values of NDVI for Bhopal city was mostly below 0.5 (S4 Table) which suggests sparse vegetation is connected with the built-up areas [83]. This alternatively explains the association of most of the dengue incidences with the high built-up areas in Bhopal. LULC classification of Bhopal shows an increase of 9.1% in built-up areas in 2020 when compared to 2014 (Table 2), and is in agreement with an earlier report [84]. Agentbased modeling shows Bhopal has grown rapidly from 1973 to 2014, with an intensification of urbanization from the city's center to the periphery, and a predicted increase in built-up areas by 240-245% between 2014 and 2022 [84]. In Bhopal, agricultural land has decreased by 45%, while the built-up areas have increased by ~260% between 1972 and 2016 pointing to rapid urbanization [85]. An increase in the built-up area was also evident from the extracted builtup area map for the period 2015–2019 (S8 Fig). In Bhopal, apart from the core urban areas, most of the urban area expansion occurred in the southern part of the city since 2010 [86]. A statistically significant dengue hotspot towards the southern part of the city (Fig 3) also signifies the influence of increasing urbanization on dengue incidences in Bhopal city. The slightly negative correlation ( $r^2 = -0.093$ ) between the association of built-up area with dengue incidence might be because of the relatively longer time period required to capture the changes in the built-up areas. Future studies should explore decadal trends to assess the spatio-temporal association of increase in built-up areas to dengue incidence. Furthermore, other socio-demographic factors, such as the increase in population density in comparison to built-up areas are also important to assess the role of urbanization on dengue incidence. From 2000-2015, Bhopal's population and built-up areas increased by 41% and 35% respectively [86]; this increase in population density might facilitate increased human-mosquito contact and could increase dengue incidence. Rapid urbanization has resulted in intense population growth (~200%) in Bhopal during the last 3 decades [85], and could be an important contributing factor for the increase in the monthly and annual mean temperature. In Bhopal, during the study period (2012–2019), the overall annual maximum, minimum, and mean temperature have increased by 0.1 °C (Test S = 10,  $\alpha$  = 0.1), and is in line with earlier findings from Bhopal [87] and Mizoram [88]. The increase in built-up areas, along with the increase in monthly and annual mean temperature will provide suitable breeding habitats for Ae. aegypti to proliferate and transmit the dengue virus.

Even though the present study has helped understand the role of local environmental factors on dengue virus transmission in Bhopal, the study has limitations, especially related to the potential bias in the dengue incidence data. The data used in the study is only from clinically diagnosed cases from government and private hospitals, and may not represent the true burden of dengue in Bhopal. The majority of the dengue cases are mild or asymptomatic, and therefore it is very difficult to estimate the true annual burden of dengue in an endemic setting [89]. Furthermore, age, gender and socio-economic status are other important factors that influence the health-seeking behavior, and thus could bias the dengue case reporting. Correcting this reporting bias may lead to more accurate and robust estimation of the dengue burden [90]. The best approach to correct the bias is to derive a multiplication factor based on the socio-economic strata of the different wards in the study area [91]. However, in India, ward level economic datasets are not available in the public domain. Based on the 2011 census (latest available), we tried to extract indirect indicators for deriving ward level multiplication factors, but were not able to find a reliable indicator.

Overall, to the best of our knowledge, this is the first report of the effect of remotely sensed meteorological and landscape variables on dengue epidemiology at a city level in Central India. In addition to devising vector control strategies, our study methodology and findings can serve as a template for understanding and forecasting dengue transmission at micro-level in different dengue-endemic climatic zones of India.

### **Supporting information**

S1 Data. Excel spreadsheet containing, in separate sheets, the underlying numerical data for Figs 2 and 4, <u>Table 3</u>, <u>S1</u>, <u>S2</u> and <u>S7</u> Figs and <u>S2 Table</u>. (XLSX)

S1 Fig. Comparison of dengue incidence (2012–2019) in Bhopal city, Madhya Pradesh and India.

(TIF)

S2 Fig. Heat-map showing ward wise dengue case density in Bhopal city (2012–2019). (TIF)

**S3 Fig. Cross-correlation of dengue cases and weather variables at 0–30 weeks lag time.** The dotted line stands for the highest r2—values with most significant correlation coefficient (p<0.0001). (TIF)

S4 Fig. Seasonal-unadjusted versus seasonal-adjusted time series data for meteorological variables (A) average temperature (°C), (B) maximum temperature (°C), (C) absolute humidity  $(g/m^3)$  and (D) rainfall (mm). (TIF)

S5 Fig. The three-dimensional plot showing the association between weekly (A) rainfall (mm), (B) Mean temperature (°C), (C) absolute humidity ( $g/m^3$ ), (D) relative humidity (%) and relative risk (RR) of dengue at different week lags. (TIF)

**S6 Fig. Changes in Land use Land cover patterns during 2014–2020 in Bhopal city.** Land use land cover classes of the study area were visually interpreted by the authors and the map layer was generated using ArcGIS version 10.2 software as described in Methods. (TIF)

**S7 Fig.** (A) Time-series plot of monthly dengue cases with monthly average NDVI values (2012–2019), (B-C) Association of dengue incidence with different ranges of NDVI values. (TIF)

**S8 Fig.** Changes in built-up area in Bhopal city between 2015 and 2019 (Source of shapefile: <u>http://projects.datameet.org/Municipal\_Spatial\_Data/</u>). (TIF)

S1 Table. Details of the satellite images used for land use land cover (LULC) classification of Bhopal city.

(DOCX)

S2 Table. Descriptive statistics of weekly values of weather variables from 2012 to 2019 in Bhopal city.

(DOCX)

S3 Table. Cross-correlation analysis between dengue cases and monthly NDVI at the different lag periods in Bhopal City. (DOCX)

S4 Table. Changes in NDVI index in high endemic wards, low endemic wards and overall NDVI of Bhopal city from 2012–2019. (DOCX)

#### Acknowledgments

The authors sincerely thank the District Malaria Officer, Bhopal district for providing the dengue epidemiology data and Dr. N. Banerjee for his encouragement to carry out this study. We also thank Dr. P. K. Mishra, Scientist-F and Head, Molecular Biology Division, ICMR-National Institute for Research in Environmental Health, Bhopal for his critical reviews and suggestion which helped to improve the manuscript substantially.

#### **Author Contributions**

**Conceptualization:** Devojit Kumar Sarma, Manoj Kumar, Praveen Balabaskaran Nina, Anil Prakash.

Data curation: Malay Pramanik, Deepanker Das, Jigyasa Dhurve.

Formal analysis: Devojit Kumar Sarma, Karuppusamy Balasubramani, Rintu Kutum.

Funding acquisition: Devojit Kumar Sarma, Rajnarayan R. Tiwari.

- Investigation: Devojit Kumar Sarma, Praveen Balabaskaran Nina, Karuppusamy Balasubramani, Anil Prakash.
- Methodology: Devojit Kumar Sarma, Praveen Balabaskaran Nina, Karuppusamy Balasubramani, Vinod Verma, Anil Prakash.
- Project administration: Devojit Kumar Sarma, Rajnarayan R. Tiwari.

Resources: Devojit Kumar Sarma, Swasti Shubham, Vinod Verma, Anil Prakash.

- **Software:** Karuppusamy Balasubramani, Malay Pramanik, Vinod Verma, Sekar Leo George, Alangar Balasundreshwaran.
- Supervision: Anil Prakash, Rajnarayan R. Tiwari.

Validation: Devojit Kumar Sarma, Manoj Kumar, Praveen Balabaskaran Nina, Vinod Verma.

- Visualization: Manoj Kumar, Malay Pramanik, Rintu Kutum, Swasti Shubham, Deepanker Das, Manoj Kumawat.
- Writing original draft: Devojit Kumar Sarma.
- Writing review & editing: Manoj Kumar, Praveen Balabaskaran Nina, Karuppusamy Balasubramani, Malay Pramanik, Rintu Kutum, Swasti Shubham, Deepanker Das, Manoj Kumawat, Vinod Verma, Jigyasa Dhurve, Sekar Leo George, Alangar Balasundreshwaran, Anil Prakash, Rajnarayan R. Tiwari.

#### References

- 1. WHO. Dengue guidelines for diagnosis, treatment, prevention and control: new edition. Geneva: World Health Organization; 2009.
- Mutheneni SR, Morse AP, Caminade C, Upadhyayula SM. Dengue burden in India: recent trends and importance of climatic parameters. Emerging microbes & infections. 2017; 6(8):e70. <u>https://doi.org/10.1038/emi.2017.57</u> PMID: <u>28790459</u>; PubMed Central PMCID: PMC5583666.
- Zeng Z, Zhan J, Chen L, Chen H, Cheng S. Global, regional, and national dengue burden from 1990 to 2017: A systematic analysis based on the global burden of disease study 2017. EClinicalMedicine. 2021; 32:100712. <u>https://doi.org/10.1016/j.eclinm.2020.100712</u> PMID: <u>33681736</u>; PubMed Central PMCID: PMC7910667.
- Messina JP, Brady OJ, Golding N, Kraemer MU, Wint GW, Ray SE, et al. The current and future global distribution and population at risk of dengue. Nature microbiology. 2019; 4(9):1508–15. <u>https://doi.org/ 10.1038/s41564-019-0476-8</u> PMID: <u>31182801</u>
- 5. Halstead SB. Dengue. The lancet. 2007; 370(9599):1644-52.

- 6. Trout A, Baracco G, Rodriguez M, Barber J, Leal A, Radke E, et al. Locally acquired dengue-Key West, Florida, 2009–2010. Morbidity and Mortality Weekly Report. 2010; 59(19):577–81. PMID: 20489680
- Ramakrishnan S, Gelfand H, Bose P, Sehgal P, Mukherjee R. The epidemic of acute haemorrhagic fever, Calcutta, 1963: epidemiological Inquiry. The Indian journal of medical research. 1964; 52:633– 50. PMID: <u>14195504</u>
- Chaturvedi U, Nagar R. Dengue and dengue haemorrhagic fever: Indian perspective. Journal of biosciences. 2008; 33(4):429–41. <u>https://doi.org/10.1007/s12038-008-0062-3</u> PMID: <u>19208969</u>
- Gupta N, Srivastava S, Jain A, Chaturvedi UC. Dengue in India. The Indian journal of medical research. 2012; 136(3):373. PMID: <u>23041731</u>
- Dhiman RC, Pahwa S, Dhillon G, Dash AP. Climate change and threat of vector-borne diseases in India: are we prepared? Parasitology research. 2010; 106(4):763–73. <u>https://doi.org/10.1007/s00436-010-1767-4</u> PMID: 20155369
- Chakravarti A, Arora R, Luxemburger C. Fifty years of dengue in India. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2012; 106(5):273–82. <u>https://doi.org/10.1016/j.trstmh.2011.12.</u> 007 PMID: 22357401.
- Arunachalam N, Murty US, Kabilan L, Balasubramanian A, Thenmozhi V, Narahari D, et al. Studies on dengue in rural areas of Kurnool District, Andhra Pradesh, India. Journal of the American Mosquito Control Association. 2004; 20(1):87–90. PMID: <u>15088710</u>.
- Jentes ES, Lash RR, Johansson MA, Sharp TM, Henry R, Brady OJ, et al. Evidence-based risk assessment and communication: a new global dengue-risk map for travellers and clinicians. Journal of travel medicine. 2016; 23(6):taw062. <u>https://doi.org/10.1093/jtm/taw062</u> PMID: <u>27625400</u>
- Ganeshkumar P, Murhekar MV, Poornima V, Saravanakumar V, Sukumaran K, Anandaselvasankar A, et al. Dengue infection in India: A systematic review and meta-analysis. PLoS neglected tropical diseases. 2018; 12(7):e0006618. <u>https://doi.org/10.1371/journal.pntd.0006618</u> PMID: <u>30011275</u>; PubMed Central PMCID: PMC6078327.
- Murhekar MV, Kamaraj P, Kumar MS, Khan SA, Allam RR, Barde P, et al. Burden of dengue infection in India, 2017: a cross-sectional population based serosurvey. The Lancet Global Health. 2019; 7(8): e1065–e73. https://doi.org/10.1016/S2214-109X(19)30250-5 PMID: 31201130
- NVBDCP. Dengue disease: National Vector Borne Disease Control Programme 2020 [cited 2020 20 April 2020]. Available from: <u>https://nvbdcp.gov.in/WriteReadData/l892s/20217657381586769254.pdf</u>.
- Patz JA, Campbell-Lendrum D, Holloway T, Foley JA. Impact of regional climate change on human health. Nature. 2005; 438(7066):310–7. <u>https://doi.org/10.1038/nature04188</u> PMID: <u>16292302</u>.
- Hii YL, Rocklov J, Ng N, Tang CS, Pang FY, Sauerborn R. Climate variability and increase in intensity and magnitude of dengue incidence in Singapore. Global health action. 2009; 2. <u>https://doi.org/10. 3402/gha.v2i0.2036</u> PMID: <u>20052380</u>; PubMed Central PMCID: PMC2799326.
- Johansson MA, Dominici F, Glass GE. Local and global effects of climate on dengue transmission in Puerto Rico. PLoS neglected tropical diseases. 2009; 3(2):e382. <u>https://doi.org/10.1371/journal.pntd.</u> 0000382 PMID: 19221592; PubMed Central PMCID: PMC2637540.
- Parham PE, Michael E. Modeling the effects of weather and climate change on malaria transmission. Environmental health perspectives. 2010; 118(5):620–6. <u>https://doi.org/10.1289/ehp.0901256</u> PMID: 20435552; PubMed Central PMCID: PMC2866676.
- Hii YL, Zhu H, Ng N, Ng LC, Rocklöv J. Forecast of dengue incidence using temperature and rainfall. PLoS neglected tropical diseases. 2012; 6(11):e1908. <u>https://doi.org/10.1371/journal.pntd.0001908</u> PMID: <u>23209852</u>
- 22. Chowell G, Fuentes R, Olea A, Aguilera X, Nesse H, Hyman J. The basic reproduction number R0 and effectiveness of reactive interventions during dengue epidemics: the 2002 dengue outbreak in Easter Island, Chile. Mathematical biosciences and engineering: MBE. 2013; 10:1455.
- Morin CW, Comrie AC, Ernst K. Climate and dengue transmission: evidence and implications. Environmental health perspectives. 2013; 121(11–12):1264–72. <u>https://doi.org/10.1289/ehp.1306556</u> PMID: 24058050; PubMed Central PMCID: PMC3855512.
- Tjaden NB, Thomas SM, Fischer D, Beierkuhnlein C. Extrinsic Incubation Period of Dengue: Knowledge, Backlog, and Applications of Temperature Dependence. PLoS neglected tropical diseases. 2013; 7(6):e2207. <u>https://doi.org/10.1371/journal.pntd.0002207</u> PMID: <u>23826399</u>; PubMed Central PMCID: PMC3694834.
- Kakarla SG, Bhimala KR, Kadiri MR, Kumaraswamy S, Mutheneni SR. Dengue situation in India: Suitability and transmission potential model for present and projected climate change scenarios. The Science of the total environment. 2020; 739:140336. <u>https://doi.org/10.1016/j.scitotenv.2020.140336</u> PMID: 32758966.

- Song HT, Tian D, Shan CH. Modeling the effect of temperature on dengue virus transmission with periodic delay differential equations. Mathematical biosciences and engineering: MBE. 2020; 17(4):4147–64. <u>https://doi.org/10.3934/mbe.2020230</u> PMID: <u>32987573</u>.
- Lambrechts L, Paaijmans KP, Fansiri T, Carrington LB, Kramer LD, Thomas MB, et al. Impact of daily temperature fluctuations on dengue virus transmission by Aedes aegypti. Proceedings of the National Academy of Sciences of the United States of America. 2011; 108(18):7460–5. <u>https://doi.org/10.1073/ pnas.1101377108</u> PMID: <u>21502510</u>; PubMed Central PMCID: PMC3088608.
- 28. Jeelani S, Sabesan S. Aedes vector population dynamics and occurrence of dengue fever in relation to climate variables in Puducherry, South India. Int J Curr Microbiol Appl Sci. 2013; 2(12):313–22.
- Campbell KM, Lin C, lamsirithaworn S, Scott TW. The complex relationship between weather and dengue virus transmission in Thailand. The American journal of tropical medicine and hygiene. 2013; 89 (6):1066. <u>https://doi.org/10.4269/ajtmh.13-0321</u> PMID: <u>23958906</u>
- Focks DA, Haile DG, Daniels E, Mount GA. Dynamic life table model for Aedes aegypti (diptera: Culicidae): simulation results and validation. Journal of medical entomology. 1993; 30(6):1018–28. <a href="https://doi.org/10.1093/jmedent/30.6.1018">https://doi.org/10.1093/jmedent/30.6.1018</a> PMID: 8271243.
- Fuller DO, Troyo A, Beier JC. El Nino Southern Oscillation and vegetation dynamics as predictors of dengue fever cases in Costa Rica. Environmental Research Letters. 2009; 4(1):014011.
- Messina JP, Brady OJ, Pigott DM, Golding N, Kraemer MU, Scott TW, et al. The many projected futures of dengue. Nature reviews Microbiology. 2015; 13(4):230–9. <u>https://doi.org/10.1038/nrmicro3430</u> PMID: <u>25730702</u>.
- Moreno-Madriñán MJ, Crosson WL, Eisen L, Estes SM, Estes MG Jr, Hayden M, et al. Correlating remote sensing data with the abundance of pupae of the dengue virus mosquito vector, Aedes aegypti, in central Mexico. ISPRS International Journal of Geo-Information. 2014; 3(2):732–49.
- Romeo-Aznar V, Paul R, Telle O, Pascual M. Mosquito-borne transmission in urban landscapes: the missing link between vector abundance and human density. Proceedings of the Royal Society B. 2018; 285(1884):20180826. https://doi.org/10.1098/rspb.2018.0826 PMID: 30111594
- Vanwambeke SO, van Benthem BH, Khantikul N, Burghoorn-Maas C, Panart K, Oskam L, et al. Multilevel analyses of spatial and temporal determinants for dengue infection. International journal of health geographics. 2006; 5:5. <u>https://doi.org/10.1186/1476-072X-5-5</u> PMID: <u>16420702</u>; PubMed Central PMCID: PMC1373612.
- 36. Bandari KP, Raju P, Sokhi B. Application of GIS modelling for dengue fever prone area based on sociocultural and environmental factors-A case study of Delhi city zone. the International Archive of The Photogrammetry, Remote Sensing and Spatial Information Sciences Beijing China. 2008; 37:165–70.
- Telle O, Vaguet A, Yadav N, Lefebvre B, Daudé E, Paul RE, et al. The spread of dengue in an endemic urban milieu–the case of Delhi, India. PloS one. 2016; 11(1):e0146539. <u>https://doi.org/10.1371/journal.pone.0146539</u> PMID: 26808518
- Rose NH, Sylla M, Badolo A, Lutomiah J, Ayala D, Aribodor OB, et al. Climate and Urbanization Drive Mosquito Preference for Humans. Current biology: CB. 2020; 30(18):3570–9 e6. <u>https://doi.org/10.</u> 1016/j.cub.2020.06.092 PMID: 32707056; PubMed Central PMCID: PMC7511451.
- United Nations DoEaSA, Population Division. World Urbanization Prospects: The 2018 Revision (ST/ ESA/SER.A/420): United Nations New York, NY, USA; 2019. Available from: <u>https://population.un.org/</u> wup/publications/Files/WUP2018-Report.pdf.
- 40. Francisco ME, Carvajal TM, Ryo M, Nukazawa K, Amalin DM, Watanabe K. Dengue disease dynamics are modulated by the combined influences of precipitation and landscape: A machine learning approach. The Science of the total environment. 2021; 792:148406. <u>https://doi.org/10.1016/j.scitotenv. 2021.148406</u> PMID: <u>34157535</u>.
- DeGroote JP, Sugumaran R, Brend SM, Tucker BJ, Bartholomay LC. Landscape, demographic, entomological, and climatic associations with human disease incidence of West Nile virus in the state of lowa, USA. International journal of health geographics. 2008; 7:19. <u>https://doi.org/10.1186/1476-072X-7-19</u> PMID: <u>18452604</u>; PubMed Central PMCID: PMC2396613.
- Machault V, Vignolles C, Pages F, Gadiaga L, Gaye A, Sokhna C, et al. Spatial heterogeneity and temporal evolution of malaria transmission risk in Dakar, Senegal, according to remotely sensed environmental data. Malaria journal. 2010; 9:252. <u>https://doi.org/10.1186/1475-2875-9-252</u> PMID: <u>20815867</u>; PubMed Central PMCID: PMC2944340.
- 43. Murdock CC, Evans MV, McClanahan TD, Miazgowicz KL, Tesla B. Fine-scale variation in microclimate across an urban landscape shapes variation in mosquito population dynamics and the potential of Aedes albopictus to transmit arboviral disease. PLoS neglected tropical diseases. 2017; 11(5): e0005640. <u>https://doi.org/10.1371/journal.pntd.0005640</u> PMID: <u>28558030</u>; PubMed Central PMCID: PMC5466343.

- 44. Keeratikasikorn C, Bonafoni S. Urban heat island analysis over the land use zoning plan of Bangkok by means of Landsat 8 imagery. Remote Sensing. 2018; 10(3):440.
- 45. Kakarla SG, Caminade C, Mutheneni SR, Morse AP, Upadhyayula SM, Kadiri MR, et al. Lag effect of climatic variables on dengue burden in India. Epidemiology and infection. 2019; 147:e170. <u>https://doi.org/10.1017/S0950268819000608</u> PMID: <u>31063099</u>; PubMed Central PMCID: PMC6518529.
- 46. Census2011. Census of India Madhya Pradesh, SERIES-24, Available at <a href="http://www.censusindia.gov.in/2011census/dchb/2327">http://www.censusindia.gov.in/2011census/dchb/2327</a> PART B\_DCHB\_BHOPAL.pdf 2011. Available from: <a href="http://www.censusindia.gov.in/2011census/dchb/2327">http://www.censusindia.gov.in/2011census/dchb/2327</a> PART B\_DCHB\_BHOPAL.pdf
- Deen JL, Harris E, Wills B, Balmaseda A, Hammond SN, Rocha C, et al. The WHO dengue classification and case definitions: time for a reassessment. Lancet. 2006; 368(9530):170–3. <u>https://doi.org/10. 1016/S0140-6736(06)69006-5</u> PMID: <u>16829301</u>.
- Gorelick N, Hancher M, Dixon M, Ilyushchenko S, Thau D, Moore R. Google Earth Engine: Planetaryscale geospatial analysis for everyone. Remote sensing of Environment. 2017; 202:18–27.
- 49. NASA. POWER LARC Data Access Viwer, Available at <a href="https://power.larc.nasa.gov/data-access-viewer/">https://power.larc.nasa.gov/data-access-viewer/</a> accessed on 30 January 2020 2020.
- Xu HY, Fu X, Lee LK, Ma S, Goh KT, Wong J, et al. Statistical modeling reveals the effect of absolute humidity on dengue in Singapore. PLoS neglected tropical diseases. 2014; 8(5):e2805. <u>https://doi.org/ 10.1371/journal.pntd.0002805</u> PMID: <u>24786517</u>; PubMed Central PMCID: PMC4006725.
- 51. ESRI. Environmental Systems Research Institute, Redlands, CA, USA. ArcGIS for Desktop, version 10.2.2 ed2014.
- 52. Rouse JW, Haas RH, Schell JA, Deering DW. Monitoring vegetation systems in the Great Plains with ERTS. NASA special publication. 1974; 351(1974):309.
- 53. Pramanik MK. Changes and status of mangrove habitat in Ganges Delta: case study in Indian part of Sundarbans. Forest Research: Open Access. 2015; 4(3).
- 54. Buchhorn M, Lesiv M, Tsendbazar N-E, Herold M, Bertels L, Smets B. Copernicus global land cover layers—collection 2. Remote Sensing. 2020; 12(6):1044.
- 55. Gilbert RO. Statistical methods for environmental pollution monitoring: John Wiley & Sons; 1987.
- Armstrong B. Models for the relationship between ambient temperature and daily mortality. Epidemiology. 2006:624–31. <u>https://doi.org/10.1097/01.ede.0000239732.50999.8f</u> PMID: <u>17028505</u>
- Gasparrini A. Distributed Lag Linear and Non-Linear Models in R: The Package dlnm. Journal of statistical software. 2011; 43(8):1–20. PMID: 22003319; PubMed Central PMCID: PMC3191524.
- 58. Team RC. R foundation for statistical computing, Vienna, Austria. URL http://wwwR-projectorg. 2013.
- Xiang J, Hansen A, Liu Q, Liu X, Tong MX, Sun Y, et al. Association between dengue fever incidence and meteorological factors in Guangzhou, China, 2005–2014. Environmental research. 2017; 153:17– 26. https://doi.org/10.1016/j.envres.2016.11.009 PMID: 27883970
- Seah A, Aik J, Ng LC, Tam CC. The effects of maximum ambient temperature and heatwaves on dengue infections in the tropical city-state of Singapore—A time series analysis. The Science of the total environment. 2021; 775:145117. <u>https://doi.org/10.1016/j.scitotenv.2021.145117</u> PMID: <u>33618312</u>.
- Bandara K, Hyndman RJ, Bergmeir C. MSTL: A Seasonal-Trend Decomposition Algorithm for Time Series with Multiple Seasonal Patterns. arXiv preprint arXiv:210713462. 2021.
- **62.** Hyndman RJ, Khandakar Y. Automatic time series forecasting: the forecast package for R. Journal of statistical software. 2008; 27:1–22.
- Mala S, Jat MK. Implications of meteorological and physiographical parameters on dengue fever occurrences in Delhi. The Science of the total environment. 2019; 650(Pt 2):2267–83. <u>https://doi.org/10.1016/j.scitotenv.2018.09.357</u> PMID: <u>30292120</u>.
- Watts DM, Burke DS, Harrison BA, Whitmire RE, Nisalak A. Effect of temperature on the vector efficiency of Aedes aegypti for dengue 2 virus. Am J Trop Med Hyg. 1987; 36(1):143–52. <u>https://doi.org/10.4269/ajtmh.1987.36.143</u> PMID: 3812879.
- Nagao Y, Thavara U, Chitnumsup P, Tawatsin A, Chansang C, Campbell-Lendrum D. Climatic and social risk factors for Aedes infestation in rural Thailand. Tropical medicine & international health: TM & IH. 2003; 8(7):650–9. <u>https://doi.org/10.1046/j.1365-3156.2003.01075.x</u> PMID: <u>12828549</u>.
- Hii YL, Rocklov J, Wall S, Ng LC, Tang CS, Ng N. Optimal lead time for dengue forecast. PLoS neglected tropical diseases. 2012; 6(10):e1848. <u>https://doi.org/10.1371/journal.pntd.0001848</u> PMID: 23110242; PubMed Central PMCID: PMC3475667.
- Mourya DT, Gokhale, Basu A, Barde PV, Sapkal GN, Padbidri VS, et al. Horizontal and vertical transmission of dengue virus type 2 in highly and lowly susceptible strains of Aedes aegypti mosquitoes. Acta virologica. 2001; 45(2):67–71. PMID: <u>11719984</u>.

- Koh BK, Ng LC, Kita Y, Tang CS, Ang LW, Wong KY, et al. The 2005 dengue epidemic in Singapore: epidemiology, prevention and control. Annals of the Academy of Medicine, Singapore. 2008; 37 (7):538–45. PMID: <u>18695764</u>.
- Tuladhar R, Singh A, Varma A, Choudhary DK. Climatic factors influencing dengue incidence in an epidemic area of Nepal. BMC research notes. 2019; 12(1):131. <u>https://doi.org/10.1186/s13104-019-4185-4 PMID: 30867027</u>; PubMed Central PMCID: PMC6417253.
- 70. Lowe R, Gasparrini A, Van Meerbeeck CJ, Lippi CA, Mahon R, Trotman AR, et al. Nonlinear and delayed impacts of climate on dengue risk in Barbados: A modelling study. PLoS medicine. 2018; 15(7): e1002613. <u>https://doi.org/10.1371/journal.pmed.1002613</u> PMID: <u>30016319</u>; PubMed Central PMCID: PMC6049902.
- 71. Devi R, Gouda K, Lenka S. Temperature-duration-frequency analysis over Delhi and Bengaluru city in India. Theoretical and Applied Climatology. 2021:1–15.
- Jeganathan A, Andimuthu R. Temperature trends of Chennai city, India. Theoretical and applied climatology. 2013; 111(3):417–25.
- Paaijmans KP, Wandago MO, Githeko AK, Takken W. Unexpected high losses of Anopheles gambiae larvae due to rainfall. PloS one. 2007; 2(11):e1146. <u>https://doi.org/10.1371/journal.pone.0001146</u> PMID: <u>17987125</u>
- Horta MA, Bruniera R, Ker F, Catita C, Ferreira AP. Temporal relationship between environmental factors and the occurrence of dengue fever. International journal of environmental health research. 2014; 24(5):471–81. <u>https://doi.org/10.1080/09603123.2013.865713</u> PMID: <u>24383820</u>
- **75.** Koenraadt CJ, Harrington LC. Flushing effect of rain on container-inhabiting mosquitoes Aedes aegypti and Culex pipiens (Diptera: Culicidae). Journal of medical entomology. 2008; 45(1):28–35. <u>https://doi.org/10.1603/0022-2585(2008)45[28:feoroc]2.0.co;2</u> PMID: <u>18283939</u>.
- 76. Dieng H, Rahman GM, Abu Hassan A, Che Salmah MR, Satho T, Miake F, et al. The effects of simulated rainfall on immature population dynamics of Aedes albopictus and female oviposition. International journal of biometeorology. 2012; 56(1):113–20. <u>https://doi.org/10.1007/s00484-011-0402-0</u> PMID: 21267602.
- 77. Benedum CM, Seidahmed OME, Eltahir EAB, Markuzon N. Statistical modeling of the effect of rainfall flushing on dengue transmission in Singapore. PLoS neglected tropical diseases. 2018; 12(12): e0006935. <u>https://doi.org/10.1371/journal.pntd.0006935</u> PMID: <u>30521523</u>; PubMed Central PMCID: PMC6283346.
- 78. Cheong YL, Burkart K, Leitao PJ, Lakes T. Assessing weather effects on dengue disease in Malaysia. International journal of environmental research and public health. 2013; 10(12):6319–34. <u>https://doi.org/10.3390/ijerph10126319</u> PMID: 24287855; PubMed Central PMCID: PMC3881116.
- 79. Talagala T. Distributed lag nonlinear modelling approach to identify relationship between climatic factors and dengue incidence in Colombo District, Sri Lanka. Epidemiology, Biostatistics and Public Health. 2015; 12(4).
- Silva FD, Santos AMd, Corrêa RdGCF, Caldas AdJM. Temporal relationship between rainfall, temperature and occurrence of dengue cases in São Luís, Maranhão, Brazil. Ciencia & saude coletiva. 2016; 21:641–6. <u>https://doi.org/10.1590/1413-81232015212.09592015</u> PMID: <u>26910171</u>
- Gubler DJ. Dengue, urbanization and globalization: the unholy trinity of the 21st century. Tropical medicine and health. 2011; 39(4SUPPLEMENT):S3–S11.
- 82. Araujo RV, Albertini MR, Costa-da-Silva AL, Suesdek L, Franceschi NCS, Bastos NM, et al. São Paulo urban heat islands have a higher incidence of dengue than other urban areas. Brazilian Journal of Infectious Diseases. 2015; 19:146–55. <u>https://doi.org/10.1016/j.bjid.2014.10.004</u> PMID: <u>25523076</u>
- **83.** Dolean B-E, Bilaşco Ş, Petrea D, Moldovan C, Vescan I, Roşca S, et al. Evaluation of the Built-Up Area Dynamics in the First Ring of Cluj-Napoca Metropolitan Area, Romania by Semi-Automatic GIS Analysis of Landsat Satellite Images. Applied Sciences. 2020; 10(21):7722.
- 84. Aithal TB, Vinay S, Ramachandra T. Agent based modelling urban dynamics of Bhopal, India. Journal of settlements and spatial planning. 2016:1–14.
- Wadwekar M, Wadwekar A, editors. Urbanisation and Environment: A Case of Bhopal. International Conference on Urban Sustainability: Emerging Trends, Themes, Concepts & Practices (ICUS); 2018.
- Project UD. Earth Observation for Sustainable Development, City Operations Report. <u>https://wwwthegpscorg/sites/gpsc/files/eo4sd\_urban\_bhopal\_city\_operationsreport\_2\_0pdf</u>. 2019. Epub 2019.
- 87. Mishra V, Shah R, Garg A. Climate change in Madhya Pradesh: indicators, impacts and adaptation: Indian Institute of Management Ahmadabad, India; 2016.

- Karuppusamy B, Sarma DK, Lalmalsawma P, Pautu L, Karmodiya K, Nina PB. Effect of climate change and deforestation on vector borne diseases in the North-Eastern Indian state of Mizoram bordering Myanmar. The Journal of Climate Change and Health. 2021; 2:100015.
- 89. Wilder-Smith A, Rupali P. Estimating the dengue burden in India. The Lancet Global health. 2019; 7(8): e988–e9. https://doi.org/10.1016/S2214-109X(19)30249-9 PMID: 31201129.
- Beatty ME, Stone A, Fitzsimons DW, Hanna JN, Lam SK, Vong S, et al. Best practices in dengue surveillance: a report from the Asia-Pacific and Americas Dengue Prevention Boards. PLoS neglected tropical diseases. 2010; 4(11):e890. Epub 2010/11/26. <u>https://doi.org/10.1371/journal.pntd.0000890</u> PMID: 21103381; PubMed Central PMCID: PMC2982842.
- Gibbons CL, Mangen MJ, Plass D, Havelaar AH, Brooke RJ, Kramarz P, et al. Measuring underreporting and under-ascertainment in infectious disease datasets: a comparison of methods. BMC Public Health. 2014; 14:147. Epub 2014/02/13. <u>https://doi.org/10.1186/1471-2458-14-147</u> PMID: <u>24517715</u>; PubMed Central PMCID: PMC4015559.